Author's response to reviews

Title: Rationale and design of a prospective study: Cervical Dystonia Patient Registry for Observation of OnaBotulinumtoxinA Efficacy (CD PROBE)

Authors:

Joseph Jankovic MD (josephj@bcm.edu)
Charles H. Adler MD, PhD (cadler@mayo.edu)
P. David Charles MD (david.charles@vanderbilt.edu)
Cynthia Comella MD (Cynthia_Comella@rush.edu)
Mark Stacy MD (mark.stacy@duke.edu)
Marc Schwartz BS (mschwartz@mednetstudy.com)
Susan M. Sutch PharmD (sue.sutch@evidencescisols.com)
Mitchell F. Brin MD (Brin_Mitchell@Allergan.com)
Spyridon Papapetropoulos MD, PhD
(papapetropoulos_spyros@allergan.com)

Version: 2 Date: 3 November 2011

Author's response to reviews: see over
3 November 2011

Dear BMC Series Editorial Production Team,

Thank you for the opportunity to revise our manuscript to conform to your journal style. We have addressed all of your comments, which are further explained below.

Formatting changes to be made:
---------------------------------
Major revisions (we require the author to make these changes)

Please remove page 1, as this is the cover letter, and not part of the manuscript (which should start with the title page).
--The first page has been deleted.

Affiliations: Please write out affiliations in full. Eg: include institution department.
--Department affiliations were added for all authors.

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods/design, discussion.
--The abstract has been revised. The results section was deleted and the information was moved into the discussion section.

Authors' contributions: Please indicate all authors have read and approved the final manuscript.
--All authors approved the final manuscript and this statement has been added at the end of the author contribution section.

Figures: Please cite the figure as 'Figure 1', rather than 'Fig'. Please also update the figure legend accordingly.
--This change has been made. It is now referred to as Figure 1.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Author list: Please remove titles.
--Titles have been removed from the author list.

Please remove on behalf of the CD-PROBE Study Group from author list and place in acknowledgments.
--This has been removed and moved to acknowledgments.

Funding: Please move the statement of funding to the Acknowledgements section and remove it from the title page
--This has been moved.
References: the reference list should contain up to 30 authors’ names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.
--The references have been revised to conform to your journal style.

Tables: Please remove vertical lines.
--Vertical lines have been removed from the tables.

Appendix: The information in the appendix in this manuscript should not be. Please place it as part of the acknowledgments and remove the appendix title.
--The Appendix section has been deleted and the names of the CD Study Group have been moved to the acknowledgments.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
--The figure has been cropped to remove white space.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
--We have had an editor look over the manuscript one more time. We are confident that any typos have been corrected.

Please let us know if you need anything further regarding our manuscript.

Kind regards,

Joe Jankovic
Baylor College of Medicine